# TOWARDS AN EU CHARTER OF DIGITAL PATIENTS' RIGHTS IN THE AGE OF ARTIFICIAL INTELLIGENCE

Hannah van Kolfschooten | University of Amsterdam Pre-print, 1 November 2024 | Under review at <u>Digital Society</u>

E-mail for correspondence: h.b.vankolfschooten@uva.nl

#### ABSTRACT

The rapid advancement of digital health innovation, including Artificial Intelligence (AI), is transforming healthcare. The growing role the European Union (EU) plays in regulating the use of AI in healthcare renders national laws insufficient to safeguard patients from unique AI-related risks. This underscores the urgent need for the recognition of a canon of patients' rights in the scope of EU law. This paper proposes the blueprint for an *EU Charter for Digital Patients' Rights*, consolidating and adapting existing rights for patients to address these specific challenges. Traditional patients' rights such as the right to privacy and the right to informed consent are interpreted in light of the challenges posed by AI. It also proposes novel rights for patients, such as the right not to be subject to automated medical decision-making and the right to meaningful human contact. This paper highlights the EU's central role in protecting patients' rights and outlines strategies for effective implementation. This comprehensive approach aims to address the current shortcomings and enhance the legal framework governing digital health in Europe.

#### **1. INTRODUCTION**

Digital health innovation is thriving. Artificial Intelligence (AI) systems are increasingly integrated into healthcare, for instance, in diagnostic tools for skin cancer, (Krakowski et al., 2024) models predicting the disease progression of Alzheimer's disease, (Fan Yi et al., 2023) models predicting patients' response to specific antidepressants, (Poirot et al., 2024), and generative AI systems documenting notes on patient visits (Blease et al., 2024). Healthcare is currently one of the top areas for AI in the European Union (EU), as AI is expected to benefit diagnostics, treatment and care.<sup>1</sup> This rapid growth underscores the urgency of addressing the rights of patients subjected to AI in healthcare, as new technology may create novel (and unforeseen) challenges.

Indeed, experts are concerned about the risks AI poses to the rights of patients (Van Kolfschooten, 2022a). While this paper primarily examines the challenges posed by AI in healthcare, it situates these within the broader context of digital health innovation, including technologies like telemedicine and other digital advancements that shape patients' rights. Digitalisation in healthcare encompasses a spectrum of innovations that collectively present opportunities and challenges for protecting patients' rights. For example, studies show how AI systems are more likely to produce incorrect predictions for members of marginalised population groups because their data is often underrepresented in the underlying datasets (Medrano et al., 2023; Lau, 2023). Moreover, in cases where medical AI systems exhibit such adverse health outcomes (Duffourc & Gerke, 2023) or cause rights violations,<sup>2</sup> it is not always clear to patients how to protect themselves. This may be exacerbated by the lack of uniform legislation on the protection of the rights of patients in Europe. This paper examines whether the EU is in need of a consolidated legal framework on patients' rights to address these specific challenges.

While generally, the importance of the rights of patients is recognised in all EU Member States and reflected in (inter)national legislation and guidelines, a uniform legal framework at the level of the EU is currently absent (Palm et al., 2020). The patients' rights framework applicable to EU Member States consists of a patchwork of national legislation (Townend et al., 2018),

<sup>&</sup>lt;sup>1</sup> The annual private investment in AI technology in the healthcare sector increased by 108,15% between 2017 (€314.48 million) and 2022 (€654.73 million), see: Annual private investment in artificial intelligence, by focus area', NetBase Quid via Al Index Report (2023) via <ourworldindata.org> accessed 15 April 2024. NB: this does not include public investments. <sup>2</sup> An example is the violation of the right to personal data protection.

various international human rights instruments and related jurisprudence,<sup>3</sup> and EU secondary laws on health-related topics.<sup>4</sup> Specifically, when AI is deployed for healthcare, the *individual rights* that patients are entitled to in the course of their medical treatment are scattered across these frameworks, causing uncertainty about their applicability, scope, and enforceability (Mira et al., 2012). Moreover, this patchwork of patients' rights creates divergent levels of protection between European patients, which could amplify health inequity within the EU (Toebes, 2021). To this end, this paper proposes consolidation through the adoption of a special legal instrument for patients' rights in the scope of EU law.

More specifically, the growing integration of AI in healthcare may necessitate adaptation of existing rights of patients – or the creation of new rights. Against this background, this paper outlines the blueprint for an '*EU Charter for Digital Patients' Rights*' in response to the challenges posed by the age of AI. Apart from the fragmented nature of the regulatory framework, this Charter is needed because current patients' rights – as protected in national healthcare settings – are not necessarily adapted to the new algorithmic reality of healthcare. To illustrate, the few existing 'soft law' instruments on patients' rights in Europe, such as the WHO Declaration on the Promotion of the Rights of the Patient in Europe (1994), are more than twenty years old and are thus not concerned with AI-specific risks.<sup>5</sup> Besides, neither the newly adopted EU AI Act nor the Council of Europe's AI Convention, *explicitly* addresses the healthcare-specific risks AI poses to patients. The main reason for this is that both instruments apply across all sectors. The EU AI Act focuses on the safety and quality of the product, primarily stipulating rules for developers instead of rights for individuals.<sup>6</sup> The AI Convention *does* introduce individual rights – but (for now) only applies to the public sector, thus generally excluding healthcare in most EU Member States (Van Kolfschooten & Shachar, 2023).<sup>7</sup>

The importance of strengthening the rights and interests of patients in relation to new technology is recognised more broadly and has been increasingly pointed out by international organisations. In July 2023, the UN Special Rapporteur on the right to health highlighted the broader challenges of digital innovation, including AI, telemedicine, and electronic health

<sup>&</sup>lt;sup>3</sup> See for example the European Convention on Human Rights (ECHR). Also see Nys, 2001.

<sup>&</sup>lt;sup>4</sup> See for example: Directive 2011/24/EU of the European Parliament and of the Council of 9 March 2011 on the application of patients' rights in cross-border healthcare.

<sup>&</sup>lt;sup>5</sup> At the supranational level, there is the World Health Organization (WHO) Declaration on the Promotion of the Rights of the Patient in Europe (1994), and declarations by private actors such as the Active Citizens Network European Charter of Patients' Rights (2002) and the World Health Assembly (WHA) Declaration of Lisbon on the Rights of the Patient (2005).

<sup>&</sup>lt;sup>6</sup> NB: the only individual right – Article 85 Artificial Intelligence Act (EU AI Act) – is merely a procedural right and does not apply to AI medical devices. Also see Van Kolfschooten, 2023.

<sup>&</sup>lt;sup>7</sup> NB: the Court of Justice of the EU (CJEU) uses a broad interpretation of "public", which may mean that in EU Member States most of the healthrelated services in the public sector would be covered, see for example: CJEU C-168/18, Pensions-Sicherungs-Verein, para. 48).

record systems, which collectively influence patients' rights and access to care. These considerations underline the importance of addressing not only AI but the broader spectrum of digital health technologies in patient rights frameworks (Human Rights Council, 2023). In October 2023, the World Health Organization (WHO) released a report outlining the key considerations for regulation of AI in healthcare, flagging its threats to informed medical decision-making by patients, personal data protection, and discrimination (WHO, 2023). In January 2024, the OECD published a report titled "AI in Health: Huge Potentials, Huge Risks", underlining the need to address key risks of AI in health, such as unclear accountability, biased and opaque algorithms, and leaks of sensitive personal data (OECD, 2024). In the same vein, in March 2024, the Lancet Commission on Health and Human Rights published a critical report, calling for an adaption of human rights to the 21<sup>st</sup> century in response to serious negative effects on health resulting from political, social, and technological developments and shortcomings in current frameworks (Beyrer et al., 2024). This criticism resembles the broader trend of the 'digital rights movement'.<sup>8</sup>

An EU Charter for Digital Patients' Rights could ensure uniform protection against the unique risks AI presents in healthcare. In this light, this paper proposes to update 'traditional' patients' rights and evaluates the inclusion of five new iterations of rights adapted to the specific context of healthcare and digital developments. At the same time, it is acknowledged that establishing a new charter alone does not ensure effective implementation in medical practice and thus discusses potential strategies to overcome this hurdle.

The paper is structured as follows. Section 2 examines the need for patients' rights protection by assessing the implications of AI. Section 3 explores whether the algorithmisation of healthcare necessitates the *adaptation* of patients' rights. In this light, it draws inspiration from the 'digital rights movement' for the establishment of potential new rights. Subsequently, Section 4 examines whether the EU is in the best position to adopt legislation to protect the rights of patients. Subsequently, Section 5 puts forward the blueprint for an EU Charter for Digital Patients' Rights. Section 6 concludes.

<sup>&</sup>lt;sup>8</sup> See on this concept: Postigo, 2012.

#### 2. HOW AI CHALLENGES HEALTH AND PATIENTS' RIGHTS

#### 2.1 Health Risks of Technical and Contextual Biases in AI

While healthcare is often presented as the most promising use case for AI, the increasing use of disruptive technologies in healthcare — such as AI — also presents significant challenges to the protection of the health and rights of patients in Europe. For example, when AI tools assist health professionals in diagnostics, health risks can materialise due to the inherent characteristics of AI.9 A primary concern is the tendency of AI systems to exhibit technical biases that are often 'built into' the system. This issue arises, for instance, when training datasets underrepresent certain population groups, such as women, black people, or patients with rare diseases - potentially missing the correct diagnosis for individuals in these groups. Inaccuracies in dataset labeling (e.g. mislabeling benign tumours as malignant) can also misguide the AI's training and result in erroneous diagnoses (Logan et al. 2023). Furthermore, the model itself can be biased, for example, an AI tool to diagnose cardiac arrest in emergency calls could be programmed to give more weight to chest pain than shortness of breath (Blomberg et al., 2019). As chest pain is typically experienced by men suffering from cardiac arrest, and shortness of breath by women, the AI tool is more likely to correctly diagnose the male caller (Greenwood et al., 2018). Recent reports submitted to the AI Incident Database<sup>10</sup> related to technical biases include (1) the WHO's AI chatbot 'S.A.R.A.H.' (described as a 'virtual health worker'11) often provided incorrect answers, especially on the topic of mental health, (2) an analysis of AI models to predict the prognosis or course of infection in patients with Covid-19 showed that the majority produced inaccurate predictions, and (3) an AI chatbot designed to provide mental health advice to children responded inappropriately to their reports of child abuse (Wynants et al. 2020; Khawaja & Bélisle-Pipon, 2023). All of these examples exhibit significant risks to the human health of patients.

Second, AI systems may exhibit *contextual* biases. This can happen when, for example, the AI system is deployed in a significantly different clinical setting or used for a different patient population than it was originally trained for (Nakagawa et al., 2023). This can also occur when the design of the AI system excludes certain patients from using it effectively, for example, if

<sup>&</sup>lt;sup>9</sup> See on the medical liability (for physical harm) by AI-powered diagnostic systems in the Netherlands: Van Kolfschooten & Wijne, 2024.

<sup>&</sup>lt;sup>10</sup> The AI Incident Database is a project of the Responsible AI Collaborative. Anyone can submit incidents, see: <incidentdatabase.ai/>. <sup>11</sup> The WHO has marketed its chatbot worldwide see <a href="https://www.who.int/news/item/02-04-2024-who-unveils-a-digital-health-promoter-harnessing-generative-ai-for-public-health">https://www.who.int/news/item/02-04-2024-who-unveils-a-digital-health-promoter-harnessing-generative-ai-for-public-health</a>.

it requires the newest smartphone (Shaw, 2023). External factors can create further risks to patients' health status, for instance, when the radiologist over-relies on the AI tool's outcome and does not consider other relevant patient information or perhaps does not have enough training and knowledge to adequately interpret the results provided by the AI system (Evans & Snead, 2024). In other words: AI systems can generate biases in all phases of the AI lifecycle. This reinforcing loop can be described as the 'AI cycle of health inequity': existing discriminatory practices are programmed into AI systems that replicate these biases in their output, eventually resulting in health inequity – avoidable – and therefore unfair - systematic differences in the health status of different population groups.<sup>12</sup>

## 2.2 Adverse Effects of AI for Patients' Rights Protection

In turn, the health risks as described above, can infringe on the rights of patients involved (Ó Cathaoir, 2022). Patients' rights in EU Member States are protected through a fragmented, multi-layered system. While each EU Member State has its unique set of laws and policies that define and protect patients' rights, the national patients' rights frameworks are in large part made up of supranational frameworks, such as health-themed EU Directives,<sup>13</sup> the legislation of the Council of Europe of which all EU Member States are part,<sup>14</sup> international treaties such as the International Health Regulations,<sup>15</sup> and 'soft law' instruments such as professional guidelines. The exact interpretation of rights varies from one Member State to another, reflecting differences in legal systems, healthcare organisation, and cultural values. Because of these shared international norms, and the common roots in bioethical principles, it is however possible to deduct a set of 'core' patients' rights that are recognised in all EU Member States,<sup>16</sup> clustered in five categories. As seen in Figure 1 below, all EU Member States protect rights in the following categories: (1) the right to access to healthcare; (2) the right to privacy and confidentiality; (3) the right to information; (4) the right to autonomy; and (5) the right to effective remedy. The identification of core patients' rights is based on a combined reading of the various - non-legally binding - patients' rights frameworks that have been published over

<sup>&</sup>lt;sup>12</sup> See for further elaboration on the 'AI cycle of health inequity': van Kolfschooten, 2023. Also see on "cascading effects" of medical AI: Leslie et al., 2021.

<sup>&</sup>lt;sup>13</sup> For example: Directive 2011/24/EU on the application of patients' rights in cross-border healthcare.

<sup>&</sup>lt;sup>14</sup> For example: the ECHR and Convention for the Protection of Human Rights and Dignity of the Human Being with regard to the Application of Biology and Medicine: Convention on Human Rights and Biomedicine (ETS No. 164) 04/04/1997. NB: the EU also recognizes the importance of the Council of Europe in Article 6 TEU.

 $<sup>^{\</sup>scriptscriptstyle 5}$  World Health Organization. (2016). International Health Regulations (2005), 3rd ed. World Health Organization.

<sup>&</sup>lt;sup>16</sup> This is confirmed by a mapping exercise of patients' rights in 30 countries in Europe, conducted January to September 2015, see: (Townend et al., 2018).

the years: the rights enshrined in the WMA Declaration of Lisbon on the Rights of the Patient (2015)<sup>17</sup>, the Active Citizenship Network's European Charter of Patients' Rights (2002),<sup>18</sup> and the WHO Europe Declaration on the Promotion of the Rights of Patients in Europe (1994).<sup>19</sup> The scope and interpretation are determined by general human rights instruments and related case law<sup>20</sup> and health-specific supranational legislation.<sup>21</sup>





First – while AI can also improve access due to its capacity to speed up medical decisions and enable remote care – the *right to access to healthcare* is potentially infringed when AI systems, influenced by technical biases, underrepresent certain demographic groups. For example, an AI diagnostic tool biased against Hispanic and Asian women may lead to underdiagnosis, directly limiting these patients' access to necessary treatments and healthcare services (Celeste et al., 2023). This also constitutes a clear violation of the right to non-discrimination, as AI may favour certain patient populations over others, creating systematic barriers to healthcare access for marginalised groups (Takshi, 2020). The opaque functioning of AI makes it difficult to identify and correct biases. Indeed, AI exhibits new forms of discrimination not directly protected by law, such as 'proxy discrimination' – differentiating on the basis of seemingly

<sup>&</sup>lt;sup>17</sup> Adopted by the 34th WMA, Lisbon, Portugal, September/October 1981 and amended by the 47th WMA General Assembly, Bali, Indonesia, September 1995 and editorially revised by the 171st WMA Council Session, Santiago, Chile, October 2005 and reaffirmed by the 200th WMA Council Session, Oslo, Norway, April 2015.

<sup>&</sup>lt;sup>18</sup> Active Citizenship Network, 'European Charter of Patients' Rights', Basis Document, Rome, November 2002.

<sup>&</sup>lt;sup>19</sup> WHO, Regional Office for Europe, 'The Rights of Patients. Report on a WHO Consultation, Amsterdam, 28-30 March 1994', in: WHO Regional Office for Europe, in collaboration with the Health Law Section, University of Amsterdam, Promotion of the Rights of Patients in Europe, London: Kluwer Law International 1995, 13-19.

<sup>&</sup>lt;sup>20</sup> E.g. the right to public health, medical care, social security and social services in Article 5(d) of the International Convention on the Elimination of all forms of Racial Discrimination (1965).
<sup>21</sup> E.g. the right to non-discrimination with regard to nationality when seeking healthcare in other Member State in Article 4(3) of Directive 2011/24/EU

<sup>&</sup>lt;sup>21</sup> E.g. the right to non-discrimination with regard to nationality when seeking healthcare in other Member State in Article 4(3) of Directive 2011/24/EU of the European Parliament and of the Council of 9 March 2011 on the application of patients' rights in cross-border healthcare.

<sup>&</sup>lt;sup>22</sup> Figure 1 is composed and designed by the author with the use of Canva (2024).

neutral categories, which however disproportionately harms members of a protected class (e.g. postal codes) (Prince & Schwarcz, 2021; Schwarcz, 2021). For example, in the CHEZ case (Case C-83/14 at para. 56), where the State had placed electricity meters in a district that was commonly known for its Roma population at an inaccessible height, the Court of Justice of the EU (CJEU) referred to this practice as 'discrimination by association'. In the same vein, contextual biases can limit access to healthcare for certain individuals, for example when the design of the AI system poses hurdles due to limited digital skills or restricted financial resources (Finco et al. 2024; Ahmed et al., 2021).

Second, the right to privacy and confidentiality could be compromised by AI systems, due to their processing and handling of vast amounts of personal health data. By monitoring and tracking various bodily functions and physical activities, potentially combined with data from external sources, AI systems can create detailed and intimate profiles about an individual's health. This can be used to make elaborate predictions about a person's health status and development. It is questionable to what extent patients can exercise control over who can access their data – also because applicable data protection regimes such as the EU General Data Protection Regulation (GDPR) do not always require the patient's consent for processing. In any case, informed consent as a mechanism to protect the confidentiality of health data is not always effective in practice, as empirical research reveals that people often click consent without reading the conditions (Van Kolfschooten, 2022b). Moreover, the increasing complexity of AI systems and data flows, compromises the capacity of individuals to understand what they consent to. Anonymisation of personal data – another solution frequently proposed to protect privacy - is not always technically possible for health data (Ghazarian, 2021). In practice, effective data protection primarily depends on adequate enforcement – which is often missing for the GDPR (Daigle & Khan, 2020).

Third, *the right to information* is affected when AI systems function as 'black boxes': it is not clear how exactly the system arrived at a certain decision for a specific patient. This lack of transparency can impair patients' understanding of their health status or the available medical interventions. When the healthcare professional cannot provide the patients with the relevant information, this may affect the level of trust in the patient-health professional relationship (Wan Kim & Routledge, 2022). Moreover, when medical AI exhibits biases, it could provide irrelevant, inaccurate, or incomplete health information to certain patients. Finally, through its

ability to find new patterns in large datasets, AI could also generate new insights into the health status of individuals – potentially violating their right not to know (Davies, 2020).

Fourth, while AI could potentially improve autonomy because of its capacity to personalise healthcare, the increasing use of AI may also jeopardise the right to autonomy. The right to informed consent does not only play a role in relation to the processing of personal data but is also an essential part of the right to autonomy. Patients need to understand the implications of their medical decisions to provide informed consent to medical treatment. When neither the healthcare professional nor the patients can explain or understand how an AI-generated diagnosis was made, for example, due to the 'black-box' nature of the AI or limited digital skills, this may affect the level of trust in the patient-health professional relationship, which may influence the patient's decision to seek care (Durán & Jongsma, 2021). Biases in AI systems, such as AI diagnostic tools, can compromise patient autonomy by providing inaccurate or incomplete information that influences the patient's medical decisions. For example, a misdiagnosis due to biased AI may result in the patient's choice of treatments that may not be necessary or effective, thereby violating their physical and mental integrity (Lau, 2023; Agarwal et al., 2022). Indeed, it is doubtful whether patients - and health professionals alike can even refuse the use of AI systems while maintaining access to high-quality healthcare (Ploug & Holm, 2020).

Finally, the emerging use of AI in healthcare may infringe on *the right to effective remedy*. If AI systems have caused harm, either in the form of physical harm or violation of human rights, this is difficult for the affected patient to prove. In the case of physical harm, proving causality can be problematic, while for rights violations, patients do not always have (1) knowledge about the violation (e.g. in the case of illegitimate data sharing with third parties) or (2) access to the data that is necessary to prove the violation (e.g. the training dataset showing how information related to their population group was not included). The opacity of AI makes it hard to identify and prove breaches (Nunez Duffourc & Gerke 2023).

In conclusion, the increasing use of AI technologies in healthcare may undermine the protection of the health and rights of patients in Europe. At the same time, it can be concluded from the above that the codification of existing rights may not suffice to protect against the new challenges posed by AI systems, as precisely these rights are threatened by the increasing integration of AI technologies in healthcare settings. Similar challenges arise with other digital health technologies, such as telemedicine, which can exacerbate inequities due to disparities in digital access and literacy. These issues highlight the need for a comprehensive approach that addresses the interconnected challenges of AI and broader digital innovations in healthcare. The next Section explores whether the algorithmisation of healthcare necessitates the adaptation of patients' rights – or the adoption of new patients' rights – and what this could look like.

# **3. ADAPTING PATIENTS' RIGHTS TO THE AGE OF AI**

#### 3.1 Responding to Political, Social, and Technological Transformations

The human rights community has been increasingly calling for a renewal of 'the paradigm of human rights and health', responding to major technological, political, and social transformations. In March 2024, the Lancet Commission on Health and Human Rights published a critical report, calling for an adaption of the fundamental principles of human rights to the 21<sup>st</sup> century, referring to numerous shortcomings in current frameworks (Beyrer et al., 2024). First, from a political perspective, human rights enforcement has focused primarily on the relationship between sovereign states and their citizens, whereas health threats increasingly transcend borders (Weber, 2022). Additionally, the power of private actors is growing, while their actions are not always covered by human rights law. This accountability gap allows private entities, such as multinational corporations (e.g. Google and Amazon) and private healthcare providers, to influence the health sector - for example by introducing disruptive health technologies - without corresponding obligations to uphold patients' rights (Hermes et al., 2020). At the same time, efforts to place human rights at the centre of health should not focus solely on the healthcare sector, because health systems are ultimately shaped by political decisions at the state level. It is therefore essential to extend the scope of patients' rights beyond the traditional patient-health professional relationship and include both public and private actors.

Furthermore, from a social perspective, in response to the wider emergence of feminist legal theory in human rights law scholarship, more recent scholarship calls for an *intersectional* approach to health and human rights. Instead of studying the social determinants of health in 'silos' (e.g. the effects of race on health outcomes *or* the effects of gender on health outcomes), it is argued that health is determined by multi-factor and multi-level intersecting individual

characteristics (Kapilashrami & Aziz, 2020). Various individual characteristics can create varying discrimination for certain individuals or population groups. A narrow interpretation of discrimination neglects the existence of multiple systems of oppression intersecting (Crenshaw, 1989). This understanding is especially important in the context of health law and policy research, as insight into the differential impact of health policy on often highly heterogeneous population groups is necessary to develop appropriate health responses without forging unfair health outcomes (Hefti et al., 2024; Hefti 2024). To illustrate, treating a black woman only as a "woman" or only as "black", ignores the varied health disparities experienced by people with both characteristics (Crenshaw, 1989). For example, a study in Germany showed that women with low income who immigrated are more likely to experience severe somatic symptoms (Barbek et al., 2024). A new patients' rights charter should thus advocate an intersectional approach.

Finally, from a technological perspective, rapid technological developments call for a novel approach to the protection of human rights in relation to health. As explained in Section 2.2, current patients' rights are undermined by the increasing use of AI, for instance when AI exhibits new forms of discrimination not directly protected by law, and when contextual biases limit access to healthcare for certain individuals, for example, due to limited digital skills or restricted financial resources. The Lancet Commission on Health and Human Rights has also raised concerns about the implications of the emerging use of AI for human rights and health, such as the implications of the capacity of algorithms to deanonymise supposedly anonymous health data; the misuse of biometric data by governments; and increasing discrimination caused by biases affecting health outcomes of marginalised groups (Beyrer et al., 2024). They conclude that "governments should ensure that existing privacy legislation, regulation, and impact monitoring keeps pace with advances in artificial intelligence", thereby underlining "the need for these measures to be underpinned by a strong human rights framework" (Beyrer et al., 2024).

In other words, recent societal developments – including the integration of AI in healthcare – necessitate the enhancement of existing patients' rights to ensure they remain effective – or the adoption of new patients' rights (Von der Decken & Koch, 2020).

#### **3.2 The Quest for Novel Human Rights**

The call for the codification of patients' rights in response to major technological, political, and social transformations, should be placed within the context of the broader quest for 'novel' human rights. The classic human rights instruments, such as the European Convention on Human Rights (ECHR), date back to the 1950s and thus did not incorporate issues related to the digital society. Over the past few decades, new human rights have emerged to address evolving social, political, and technological challenges. While some human rights have developed in the context of the 'living instrument doctrine' - the approach taken by the European Court of Human Rights (ECtHR) to the evolutive interpretation of the rights enshrined in the ECHR in light of present-day circumstances,<sup>23</sup> other human rights have been codified as 'new' human rights.<sup>24</sup> Worldwide, human rights activists, non-governmental organisations, and scholars have advocated for legal recognition of specific new human rights (Von der Decken & Koch, 2020), for example in relation to rights of "vulnerable" groups such as women's rights,<sup>25</sup> and children's rights.<sup>26</sup>

In the 2000s, the 'digital rights movement' emerged in response to the development of new types of technologies (Murphy, 2009), for example in the field of gene editing and internet access (Bertolaso & Marcos, 2022; Pollicino, 2022). In the last years, there has been a surge in AI rights. Advocacy statements, such as the 2017 proposal by ARTICLE 19 for a 'Universal Declaration of Digital Rights,' emphasise the urgent need for a cohesive global approach to digital and AI-related human rights (#InternetOfRights, 2017). Instruments such as the Declaration on European Digital Rights and Principles (European Commission) and the AI Convention on Human Rights (Council of Europe) provide foundational frameworks for these emerging rights. Most recently, there has been a call for the adoption of 'neurorights' (Ienca & Andorno, 2017; Bublitz, 2022; Nawrot et al., 2024): new human rights specific to the challenges posed by neurotechnology: "the field of devices and procedures used to access, monitor, investigate, assess, manipulate, and/or emulate the structure and function of the neural systems of animals or human beings" (UNESCO, 2023). In response, UNESCO is currently drafting an international instrument on the protection of neurorights (UNESCO, 2024).

<sup>&</sup>lt;sup>23</sup> First adopted in Tyrer v. the United Kingdom App no 58/56/72 (ECtHR, 25 April 1978) para 31.

<sup>&</sup>lt;sup>24</sup> For example, the human right to a clean, healthy, and sustainable environment by the UN. See: Bob, 2011.

<sup>&</sup>lt;sup>25</sup> Codified by the United Nations (UN) in: Convertion on the Elimination of All Forms of Discrimination against Women (adopted 18 December 1979, entered into force 3 September 1981) 1249 UNTS 1. <sup>26</sup> Codified by the UN in: Convention on the Rights of the Child (adopted 20 November 1989, entered into force 2 September 1990) 1577 UNTS 3.

However, none of these instruments are specific to the context of healthcare and thus provide direct protection to the rights of patients. While the core patients' rights are generally protected under human rights law, it is important to articulate specific rights for patients and contribute to the visualisation of harm occurring in this context. In this way, States' responsibilities towards protection are promoted, and national courts and human rights bodies can apply the rights of patients to specific contexts. As concluded before in Section 2.2, current patients' rights do not specifically protect against the increasing use of AI. This seems to leave a gap in the protection of individuals in the healthcare sector (Van Kolfschooten, 2022a). In this light, it may be necessary to add new patients' rights to modernise the current catalogue of European patients' rights.

# 3.3 Time for New Patients' Rights?

As explained in Section 2.2 above, it is possible to deduct fourteen 'core' patients' rights clustered in five categories across all EU Member States from various supranational instruments, human rights legislation and accompanying case law, and 'soft law' instruments such as professional guidelines. However, as shown before, these rights do not necessarily protect patients from the new challenges arising from AI systems – at least in their traditional interpretation. Upon analysing the key scholarship on new digital rights (Custers, 2022; Nawrot et al., 2024; Ienca & Andorno 2017; Hertz, 2022; Von Arnauld et al., 2020), five rights were identified that have been previously proposed in the literature in relation to other areas that could qualify as new patients' rights:<sup>27</sup> (1) The right not to be subject to automated medical decision-making; (2) The right to meaningful human contact; (3) The right to mental privacy; (4) The right to digital health literacy; (5) A right to access new healthcare technologies. These rights respond to common threats, are based on important social values, and are capable of achieving consensus in the EU (Alston, 1984).<sup>28</sup> While some of these rights should be considered individual rights - and can be exercised only individually - others should be seen as collective rights. For example, given the collective interest it serves, the right to digital health literacy is a collective entitlement to all members of the patient community. This means that a violation affects all members of the group. The right to access new health technologies has both an individual – access to technology X in the treatment of patient Y – and a collective

 <sup>&</sup>lt;sup>27</sup> Such as AI rights; neurorights; internet rights.
 <sup>28</sup> In accordance with the criteria for new *international* human rights as argued in: (Alston, 1984).

component, e.g. inclusion of technology Z in the basic health insurance of all patients. The same could apply to the right to mental privacy, because apart from individual privacy harms, the large training datasets for AI allow for group profiling of a group of patients sharing similar characteristics.<sup>29</sup>

## 3.3.1 The right not to be subject to automated medical decision-making

Ethicists have argued that a right to not be subject to automated medical decision-making could contribute to the protection of health privacy as protected in Article 7 of the EU Charter of Fundamental Rights (CFR) (Ploug & Holm, 2020). Such a right entails that - under certain circumstances - patients should have the right to refuse medical procedures based on decisions taken with the aid of assisting AI (e.g. diagnostics or treatment selection) and from the use of partially and fully automated decision-making (e.g. AI cardiac monitoring or precision medicines) as part of their individual medical treatment while having access to a non- or lessautomated alternative treatment (Ploug & Holm, 2020). The patients' right not to be subject to automated medical decision-making ensures human involvement in medical care. As AI becomes more integrated into healthcare, decisions made by algorithms can lack the nuance and empathy that human professionals provide (Montemayor et al., 2022). This right protects patients from potential errors, biases, and dehumanisation associated with AI systems. It emphasises the need for human judgment in critical health decisions, safeguarding patient autonomy, personalised care, and the ethical use of AI in medicine. Such a right is currently absent in the EU patients' rights framework, although a health-conformant reading of Article 22 GDPR could provide patients with an enforceable right in this respect (Van Kolfschooten, 2024b).

## 3.3.2 The right to meaningful human contact

The Rathenau Institute also suggested the adoption of the *right to meaningful human contact* in relation to the notion of human dignity – the basis of human rights enshrined in the CFR. It encompasses the need for genuine interpersonal interactions, which is crucial for emotional and social well-being (Ratheneau Instituut, 2017). This could be relevant to include as a patients' right in the Charter. The right to meaningful human contact is frequently brought up in debates

<sup>&</sup>lt;sup>29</sup> For an extensive explanation of the difference between individual and collective rights, see: (Chalabi, 2023) See on privacy as a collective right: (Puri, 2023)

about using social or care robots in assisted living and elderly care situations (Laitinen et al., 2016) solitary confinement in prisons (Naylor & Shalev, 2022), and isolation of patients in psychiatric hospitals (Brown et al., 2020) Human contact is one of the fundamental aspects of healthcare. There is a widespread fear that the growing use of (AI) technology will eventually replace human contact and increase loneliness among vulnerable groups (European Parliament, 2017). As AI is becoming more common in healthcare – and may perhaps reach the extent of full automation in some areas – recognising this right as a patients' right is increasingly important. It could be recognised as part of the right to respect for private life in Articles 7 CFR and 8 ECHR.

#### 3.3.3 The right to mental privacy

Scholars in the field of 'neurorights' have been advocating a right to mental privacy as extension of Articles 7 and 8 CFR: a right to control their own neural or brain data (Ienca & Andorno, 2017; Andorno & Lavazza, 2023; Lighart et al., 2022). In the context of healthcare, brain data has been used for a long time for diagnostic purposes (for example with EEG systems). Recent developments in technology allow for increasing collection, processing, and use of brain data (potentially 'mind reading') and direct interaction with neural processes (potentially 'brain alteration'), which could lead to intrusion of the personal sphere, potentially compromising the freedom of thought (Shen, 2013). While there is thorough legal protection for other types of information about the body, brain information as such is currently not considered personal data, thus excluding it from the sphere of privacy and data protection rights (Ienca & Andorno, 2017). Encompassing a patients' right to mental privacy in healthcare is critical for protecting patients' thoughts, emotions, and cognitive processes from intrusive technologies. With the risk of AI analysing mental states, there's a risk of unauthorised access to intimate mental data, potentially leading to misuse, discrimination, or manipulation. Safeguarding mental privacy as part of the rights to respect for private life and personal data protection ensures that patients retain control over their cognitive space, fostering trust in healthcare systems and preventing psychological harm (López-Silva et al., 2024).

## 3.3.4 The right to digital health literacy

The *right to digital health literacy* ensures individuals have the knowledge and skills to understand explanations of the functioning and instructions on how to engage with digital health

technologies, including AI. This right is crucial as healthcare increasingly relies on digital tools for diagnosis, treatment, and management. Patients need to comprehend how these technologies work, their benefits, and potential risks to make informed decisions about their health. Promoting digital health literacy fosters trust, enhances patient autonomy, and ensures equitable access to healthcare, preventing disparities in the use of advanced medical technologies. It has been described as a "super social determinant of health" (Kickbusch et al., 2021). The AI transformation of the healthcare sector urges the need for patients to develop a high level of *AI health literacy*. AI health literacy is of paramount importance to establish *patients' trust* in healthcare professionals and the healthcare system when AI medical decision-making becomes the standard (Rodríguez et al., 2013; Tsai et al., 2018). AI health literacy must not be seen as the responsibility of the patient, but rather be regarded as a key enabler of patients' rights protection, therefore demanding adequate safeguarding in the EU legal order. The other side of the coin is that healthcare professionals should have the obligation to have a sufficient level of AI literacy when deploying AI systems.<sup>30</sup> The patients' right to digital health literacy could be acknowledged in relation to the right to informed consent (Ali et al., 2024).

#### 3.3.5 The right to access new healthcare technologies

In light of the emerging development of technology for medical treatment, including AI systems, the right to access to healthcare could encompass a patients' *right to access new healthcare technologies*. Generally, the right to access to healthcare is not interpreted as a right to a specific treatment, but rather as an obligation to ensure access to essential medicines and equitable distribution of health goods.<sup>31</sup> Patients have a right to access to healthcare which is available to the population generally.<sup>32</sup> Moreover, States do have an obligation to the progressive realisation of the right to health, meaning they need to invest in improving healthcare (Ó Cathaoir, 2022).<sup>33</sup> While the adoption of a right to access new healthcare technologies would not include a right to access to a specific technology tool – also because this would be dependent on financial resources (Calcoen et al., n.d.), it does entitle patients to the enjoyment of the benefits of the healthcare technologies that are generally available for the wider population. In this sense, it is comparable to Article 15 of the International Covenant on

<sup>&</sup>lt;sup>30</sup> This is protected in Article 4 of the AI Act. However, this provision addresses "providers and deployers" of AI systems, so only applies to the State when deploying AI.

 <sup>&</sup>lt;sup>31</sup> See Asiye Genç v. Turkey App no 24109/07 (European Court of Human Rights (ECtHR), 27 January 2015) para 82.
 <sup>32</sup> See Nitecki v. Poland App no 65653/01 (ECtHR, 21 March 2002) para 4.

<sup>&</sup>lt;sup>33</sup> Also see Report of the Special Rapporteur on the right of everyone to the enjoyment of the highest attainable standard of physical and mental health, A/HRC/29/33, UN Human Rights Council, 2 April 2015, para 19-22.

Economic, Social and Cultural Rights (ICESCR), which protects "the right of everyone to enjoy the benefits of scientific progress and its applications" (Ó Cathaoir, 2022).

# 4. PROTECTING PATIENTS' RIGHTS AT THE EU LEVEL

#### 4.1 Limited Competencies, Yet Significant Impact on Healthcare

While the EU holds a key position in the regulation of the internal market, therefore facilitating the availability of medical AI technologies in the EU market, it does not provide specific rights for the patients subjected to AI. Indeed, a consolidated EU framework for the protection of the rights of patients does not yet exist. Patients' rights protection in EU Member States is characterised by a patchwork of laws in a multi-level legal framework. The codification of patients' rights is not merely a symbolic but a practical tool for improving healthcare systems, protecting human dignity, and ensuring equitable treatment of patients because of its role in education and awareness, the impact on national policies, the advantages for identifying violations, and patient empowerment (Cohen & Ezer, 2013).

It is often argued that the EU's possibilities to make legislation in the area of healthcare are limited, as the organization and delivery of health services and medical care is the primary responsibility of the Member States, and harmonisation is constitutionally restricted (Greer et al., 2022). In light of the principles of conferral and subsidiarity, the EU's power in healthcare is limited to taking supporting measures (Greer et al., 2022b; Sauter & Guy, 2017). The Treaty on the Functioning of the European Union (TFEU) does confer some explicit competencies in the realm of health upon the EU in Article 168, such as measures setting high standards of quality and safety for medicinal products and devices for medical use. However, because of its competencies concerning the regulation of the internal market, the body of EU health law and policy deriving from internal market legislation is growing (Purnhagen et al., 2020). The principle of free movement of goods allows for health products and services to freely circulate in the EU. With European cooperation in cross-border healthcare growing, concomitant EU rules on health products and services, intended to promote and protect the internal market, are increasing (Roscam Abbing, 2010). In the Tobacco Advertising-case, the Court of Justice of the EU (CJEU) ruled that, while the EU's harmonizing measures may constitute the side effect of protecting human health, it is not allowed to create regulation on the basis of other Treaty

articles with the sole objective of health protection. Therefore, in practice, EU health law is often a by-product of internal market regulation on the basis of Article 114 TFEU (De Ruijter, 2016). This development shows that EU health law and policy are mainly grounded in economic considerations and do not specifically consider health or patients (Greer et al., 2022b).

Indeed, many EU (health) laws and policies made on the basis of the internal market competence influence patients' rights (Nys & Goffin, 2011). Through the regulation of goods, services, and people related to healthcare – such as AI systems,<sup>34</sup> personal data,<sup>35</sup> medical devices and medicines,<sup>36</sup> the EU already enters the realm of health law and patients' rights. Indeed, while the EU has limited direct power in shaping health policies in the Member States due to a lack of competencies, it does co-create effects for patients' rights through broader legislation and policies, including the regulation of AI systems. By designing the conditions for AI systems to enter the EU market, the EU also determines the extent to which patients may encounter the potential harm to their health or rights exhibited by these AI systems (van Kolfschooten, 2022a, 2023). From this point of view, it could be argued that the EU should, by allowing these risks, also provide concomitant protection for patients' rights within the scope of EU law. The EU's responsibilities for the protection of human health and human rights may even demand such protection (De Witte, 2006).

# 4.2 EU Responsibilities for Health Mainstreaming and Protecting Human Rights

While all EU Member States do have protections in place for the rights of patients (Townend et al. 2018), at the EU level there is currently no corresponding instrument that provides uniform recognition of the rights of patients. In light of the expanding impact of the EU on healthcare, this constitutes a gap in protection for European patients when applying and interpreting EU law. One of the key objectives of the EU is to "promote the well-being of its peoples" (Article 3 TEU). This objective is accompanied by a so-called "health mainstreaming" obligation. "A high level of human health protection shall be ensured in the definition and implementation of all Union policies and activities" (Articles 9 and 168(1) TFEU; Article 35 CFR). When adopting legislation to harmonise the internal market, the European Commission needs to take as a base

<sup>&</sup>lt;sup>34</sup> In the EU AI Act.

<sup>&</sup>lt;sup>35</sup> For example, through the General Data Protection Regulation (GDPR), the European Health Data Space, and the ePrivacy regulation.
<sup>36</sup> In the Medical Devices Regulations and legislation on clinical trials and pharmaceutical products.

a high level of protection of human health (Article 114(3) TFEU). This means that, while the competency of the EU to adopt legislation directly determining national health policies is restricted, the EU does have to safeguard human health in its legislation applicable in other areas, including internal market legislation on health technology (Hervey & McHale, 2015). The EU's commitment to health mainstreaming shows its awareness of the potential (undesired or unexpected) effects of its laws and policies on health. Technically, the provision on health mainstreaming could be relied on to invalidate EU legislation that does not provide a sufficiently high level of protection. This however seems difficult to enforce in practice (Bartlett, 2024).

Second, the EU has obligations toward the protection of human rights – also those related to healthcare. Respect for human rights is among the EU's founding values (Article 2 TEU). To a certain extent, the intrinsic connection between health and human rights – advocated in the 'health and human rights' scholarship (Toebes et al., 2022; Gostin 2020; McHale & Hervey, 2015; Murphy, 2013; Mann et al., 1999) – is recognised in the EU legal order (Herrmann et al., 2022). For instance, the importance of health in relation to human rights is also enshrined in Article 35 CFR, protecting the right to "access to preventive health care and the right to benefit from medical treatment under the conditions established by national laws and practices". It is recognised in the CFR that EU law and policy that serve the interest of health protection may under certain circumstances limit other human rights. Also, for example, the framework on anti-tobacco legislation and restrictions of the fundamental principles of the internal market, shows how health-related law and policy (e.g. health warnings on tobacco packaging) can sometimes limit other rights and interests (e.g. free movement of goods) (Delhomme, 2017).

Moreover, in light of increasing regulatory complexity due to (1) the expansion of EU action in an increasing number of areas (indirectly) affecting national health laws and policies, (2) a growing number of international treaties governing health rights to which all EU Member States are parties,<sup>37</sup> and (3) growing digitalization creating new questions and challenges for protecting patients' rights (Legemaate, 2024), the EU is the most obvious choice for locating a new patients' rights instrument. Exhaustive harmonisation of a specific policy area prevents Member States from adopting national measures deviating from EU legislation.<sup>38</sup> The AI Act

 <sup>&</sup>lt;sup>37</sup> E.g. the UN Convention on the Rights of the Child (CRC) and the UN Convention on the Rights of Persons with Disabilities (CRPD). The EU ratified the CRPD in 2011.
 <sup>38</sup> See Article 114 Treaty on the Functioning of the EU (TFEU) and see, for example, Court of Justice of the European Union (CJEU), L.E.G.O (C-242/17 L.E.G.O) ECLI:EU:C:2018:804, 4 October 2018, para 52.

aims to harmonise the entire area of AI systems ("the placing on the market, the putting into service, and the use of AI systems").<sup>39</sup> This means that, generally, Member States cannot deny market access to products that meet the requirements of the AI Act or stipulate additional rules that may frustrate the internal market.<sup>40</sup> While the AI Act does have the objective to protect human rights, especially with regard to 'high-risk' AI systems, these rules are insufficient to protect the rights of patients for multiple reasons. First, the instrument applies across all sectors, thereby disregarding the specific issues in healthcare. Second, it focuses on the safety and quality of the product rather than on rights for individuals (Van Kolfschooten, 2022a). Finally, it contains many exceptions precisely for medical devices using AI (Van Kolfschooten & Van Oirschot, 2024). Thus, while the EU Member States have the primary responsibility for healthcare and health policy,<sup>41</sup> due to the approach of maximum harmonisation to AI systems and other (health) technologies,<sup>42</sup> the EU seems to be better positioned to adopt legislation to protect the rights of patients in relation to those measures.

# 4.3 Towards A Uniform EU Approach to Patients' Rights Protection

The increasing use of AI in healthcare exacerbates the need for a unified legal approach to patients' rights within the EU in the form of an EU Charter for Digital Patients' Rights. The consolidation and modernisation of the catalogue of European patients' rights in one EU legal instrument has multiple advantages. First, a codified framework would empower patients by clearly defining their rights, enabling them to advocate more effectively and seek legal redress when necessary. This is necessary to protect patients against the undesirable actions of health professionals when deploying AI in healthcare in an inherently asymmetric relationship. Codification ensures that both patients and healthcare professionals understand their rights and obligations, allowing for more consistent interpretation, potentially preventing disputes, and strengthening judicial oversight (London, 2008). Second, codifying patients' rights can improve public health outcomes by ensuring that patients have consistent access to high-quality healthcare and that their rights are protected, thereby improving overall trust in the healthcare system (Bærøe et al., 2020). Third, it would align the interpretation of patients' rights throughout EU Member States, ensuring equal rights and protections, and reducing disparities

<sup>39</sup> See Article 1(2)(a) EU AI Act.

<sup>&</sup>lt;sup>40</sup> NB: Member States can only adopt legislation and policy posing barriers to the internal market to protect human health if this is justified, see Articles 49 and 56 TFEU. Also see: Veale & Zuiderveen Borgesius, 2021.

 <sup>&</sup>lt;sup>41</sup> Article 168(7) TFEU.
 <sup>42</sup> Also see the Medical Devices Regulation.

across the EU. This could lead to uniform enforcement mechanisms across the EU, providing more effective protection and redress for patients whose rights are violated. Finally, a codified set of patients' rights simplifies the current complex and fragmented legal landscape when AI infringes on rights, providing a clear and consistent legal framework that is easier to navigate. This is warranted by the sensitive nature of the patient-health professional relationship and the patient's position of vulnerability and dependency when in need of healthcare.

New rights could extend protection to individual patients and patient groups, that were previously unrecognised under existing human rights frameworks, which allows for a better response to contemporary issues in healthcare. Legal recognition of patients' rights provides a uniform, clear, and effective framework, leading to better implementation and accountability. An EU Charter for Digital Patients' Rights is a logical step in light of political, social, and technological developments (Beyrer et al., 2024). Such an initiative would align with recent efforts to intensify the 'European Health Union' – creating more collaboration in the field of health in reaction to the Covid-19 pandemic – and the wish to protect human rights in relation to AI (Alemanno, 2020). The next Section proposes the blueprint of such an instrument.

#### 5. PROPOSAL FOR AN EU CHARTER OF DIGITAL PATIENTS' RIGHTS

#### 5.1 Reasons for and objectives of the proposal

Major technological, political, and social transformations of society affect the human rights of individuals in Europe. Generally, the intensifying cross-border character of healthcare due to increasing international pandemics, the integration of non-state private actors such as multinational technology companies, and the increasing online, non-territory bound character of healthcare, necessitates additional EU action in this field. Especially the increasing algorithmisation of healthcare – while introducing considerable opportunities to improve the accessibility, efficiency, and quality of healthcare – presents significant challenges to the rights of patients. The responsibilities the EU has towards the protection of human rights (Article 2 TEU), in combination with the ongoing expansion of EU policies directly influencing the rights of patients,<sup>43</sup> call for a stronger protection of patients' rights in the scope of EU law.

<sup>&</sup>lt;sup>43</sup> For example on topics such as pharmaceutical products; medical devices; clinical trials; serious cross-border threats to health; patients' rights in cross-border healthcare; AI systems; personal data; health data.

Depending on the exact content the Charter, the legal basis for this proposal could be Articles 114, 168, 16 and 19 TFEU. Article 114 TFEU concerns the internal market and allows for the adoption of measures necessary for the establishment or functioning of the internal market, whilst at the same time ensuring a high level of public health protection as required in Article 114(3) and 168(1) TFEU, and fully respecting the responsibilities of the Member States to organise, finance and deliver health services and medical care. Article 16 TFEU allows for the adoption of rules for the protection of individuals with regard to the processing of personal data and Article 19 TFEU allows for measures to combat discrimination. The underlying idea is that, when taking measures to protect the internal market, the EU ought to protect the health – and rights – of patients too.

The proposal concerns the protection of the rights of patients and is an important building block of the European Health Union (European Commission, 2024). It sets out the rights and freedoms individuals are entitled to in the context of healthcare technologies. It aims to address the fragmented approach to the protection of the rights of patients in Europe, caused partially by the complex regulatory framework of EU secondary law related to healthcare and health-related products, and exacerbated by new technological, political, and social developments. The *EU Charter of Digital Patients' Rights* responds to these developments. It departs from the understanding that patients' rights have their basis in human rights and thus warrant thorough protection. It recognises that various individual characteristics can create varying discrimination for certain individuals or population groups, and therefore emphasises the need for an intersectional approach to the rights enshrined in this Charter.

Against this background, the objective of this proposal is twofold. First, it aims to clarify how EU human rights and values should be applied in the context of healthcare. Second, it translates these human rights and values to apply in the new digital reality. In doing so, it builds on the EU Charter of Fundamental Rights and the case law of the Court of Justice of the EU and the European Court of Human Rights. The ECtHR's case law is relevant because of Article 52(3) TFEU: the meaning and scope of the rights in the CFR are the same as the corresponding rights in the ECHR. Explicating the rights of patients strengthens protection by making those rights more visible in one Charter. Ultimately, it promotes the rights enshrined in the Charter in order to strengthen the position of patients and guide relevant stakeholders in the implementation

(e.g. States; individual healthcare professionals; hospitals; and technology developers). The Charter sets out in a single text all the rights of patients receiving healthcare in the EU.

# 5.2 Proposed Scope of Application

The Charter protects the rights of European patients in relation to the areas subjected to EU harmonisation of health, such as health technology. The rights and entitlements in the Charter shall apply to individuals receiving healthcare, or seeking to receive healthcare in an EU Member State, regardless of how it is organised, delivered and financed. With 'healthcare' is meant "health services provided by health professionals to patients to assess, maintain or restore their state of health, including the prescription, dispensation and provision of medicinal products and medical devices".<sup>44</sup> Because many of the rights enshrined in the proposed Charter have their basis in the CFR, it is logical to follow the same scope of application. The Charter applies to the institutions, bodies, offices and agencies of the Union and to the Member States only when they are implementing Union law.<sup>45</sup> Because of the expanding EU action in areas related to health and healthcare, such as (AI) medical devices, personal (health) data, and clinical trials, this scope is broad enough to cover the main challenges for patients' rights.

While Article 168(7) TFEU refrains the EU from determining national health policies and laws, and EU patients' rights could thus only be applied within the scope of the implementation of EU law, they could still have a *de facto* impact above and beyond EU law by influencing EU Member States to implement these rights in other areas.

#### 5.3 Legal Status and Choice of the Instrument

Human rights scholars have recognised numerous pathways to the adoption of new human rights - in this case, patients' rights. In the derivation approach, new human rights can be derived from existing rights through interpretation, for example through the living instrument

<sup>&</sup>lt;sup>44</sup> Definition of healthcare used in Directive 2011/24/EU of the European Parliament and of the Council of 9 March 2011 on the application of patients' rights in cross-border healthcare (Article 3(a)). <sup>45</sup> See Article 51(1) Charter of Fundamental Rights of the European Union (CFR).

doctrine by courts to address new situations that fall within the scope of existing rights. In the *treaty approach*, new human rights are established through the adoption of new treaties or optional protocols to existing treaties. Another pathway is the *soft law approach*, where new human rights are first mentioned in non-binding resolutions and declarations by international organisations. Over time, these soft law instruments can pave the way for formal recognition and the drafting of binding treaties (Von der Decken & Koch, 2020).

The EU Charter of Fundamental Rights has the form of an EU Treaty, signed and ratified by all EU Member States. This proposal does not advocate for a treaty revision or adoption of an additional protocol: amendment of the CFR would require unanimous approval of all Member States and is thus a politically complicated and time-consuming procedure. The EU can act instruments: Directives, through enumerated legal Decisions, Regulations, and Recommendations and Opinions (Article 288 TFEU). Directives and Decisions are not desirable because of the relative freedom they provide for implementation by Member States potentially leading to even more fragmentation. The form of the Regulation seems to be the best option in the long term – given the need for a regulatory framework that directly addresses the rights of patients and reduces fragmentation and inconsistency in the digital single market of healthcare.

However, in light of the current political climate in Europe, for now, this proposal could be adopted in the form of a Recommendation, for example by the Council of the European Union. The Council Recommendation does not constitute a binding act, but it sets certain standards that EU Member States are expected to oblige. It thus counts on the moral authority of the EU. In this way, this proposal takes the *soft law approach*, by first listing the rights of patients in a non-binding instrument. Over time, this soft law instrument can pave the way for formal recognition and the drafting of binding legislation, such as a Regulation (Von der Decken & Koch, 2020). In the meantime, stakeholders can use the Charter to guide their national policies. Because of the embedding of the proposed patients' rights in current human rights law, national courts and the CJEU can use the *derivation approach* and treat the Charter as an interpretative tool. A shortcoming of the Council Recommendation is the lack of democratic underpinning by the European Parliament. Another option could thus be to adopt a Declaration<sup>46</sup> – an instrument that is becoming more popular in the 'digital sphere'. However, this instrument has no binding

<sup>&</sup>lt;sup>46</sup> NB: the EU institutions can issue a Declaration to convey their point of view. This is however not an official legislative instrument.

power, does not introduce new obligations or entitlements, and should merely be seen as statements of intention for future EU strategies. At the same time, the fact that Declarations are signed by (often multiple) key institutions of the EU contributes to their value (Cocito & De Hert, 2023).

# 5.4 Proposed Catalogue of European Patients' Rights

Category	Right	Legal Basis
(1) The right to access to	The right to availability,	Article 35 CFR
healthcare	accessibility, acceptability,	• Article 2 and 8 ECHR
	and quality of healthcare	• Article 11 European Social
		Charter
		• Article 25 Universal
		Declaration of Human Rights
		Article 12 ICESCR
		• Article 3 Convention on
		Human Rights and
		Biomedicine
	The right to non-	Article 14 ECHR
	discrimination	• Article 4(3) Patients' Rights
		Directive
		• Article 11 Convention on
		Human Rights and
		Biomedicine
	* The right to access new	Article 15 ICESCR
	healthcare technologies	• Article 2 ECHR
(2) The right to privacy and	The right to privacy	• Article 7 CFR
confidentiality		• Article 8 ECHR

		• Article 10(1) Convention
		on Human Rights and
		Biomedicine
		• Article 4.7 and 4.8 WHO
		Europe Declaration on the
		Promotion of the Rights of
		Patients in Europe (1994)
	The right to medical	• Article 4.1 and 4.2 WHO
	confidentiality	Europe Declaration on the
		Promotion of the Rights of
		Patients in Europe (1994)
	The right to medical data	Article 8 CFR
	protection	• Article 9 GDPR
		• Article 4.3 WHO Europe
		Declaration on the Promotion
		of the Rights of Patients in
		Europe (1994)
	* The right to mental privacy	Article 7 and 8 CFR
		Article 8 ECHR
		• Neurorights proposals
		(UNESCO)
		Article 9 GDPR
(3) The right to information	The right to information on	Article 8 ECHR
	one's health	• Article 10(1) Convention
		on Human Rights and
		Biomedicine
		• Article 4(2)(f) Patients'
		Rights Directive
		• Article 12(5), 15(1) and
		15(3) GDPR (Access to
		medical records)

		• CJEU, C-307/22 ( <i>FT v</i> .
		DW) (Access to medical
		records)
	The right to information on	Article 8 ECHR
	treatment options	• Article 5(2) Convention on
		Human Rights and
		Biomedicine
		• Article 2.2 WHO Europe
		Declaration on the Promotion
		of the Rights of Patients in
		Europe (1994)
		• Article 4(2)(b) Patients'
		Rights Directive
	The right not to know	Article 8 ECHR
		• Article 10(2) Convention
		on Human Rights and
		Biomedicine
	* The right to digital health	Article 8 ECHR
	literacy	Article 12 ICESCR
		• Article 5 Convention on
		Human Rights and
		Biomedicine
(4) The right to autonomy	The right to informed	• Article 3(2) CFR
	consent	• Article 3 and 8 ECHR
		• Article 5 Convention on
		Human Rights and
		Biomedicine
		Article 7 ICCPR
		• Article 9 GDPR
	The right to physical	• Article 3(1) and 4 CFR
	integrity	• Article 3 and 8 ECHR
		• Article 1.2
	The right to mental integrity	• Article 3(1) and 4 CFR

		• Article 3 and 8 ECHR
		• Article 1.2 WHO Europe
		Declaration on the Promotion
		of the Rights of Patients in
		Europe (1994)
	* The right to meaningful	• Article 1 CFR
	human contact	• Article 8 ECHR
	* The right not to be subject	Article 8 ECHR
	to automated medical	• Article 22 GDPR
	decision-making	• CJEU, Case C-634/21,
		SCHUFA Holding (Scoring)
(5) The right to effective	The right to post-treatment	Articles 6 and 8 ECHR
remedy	information	• ECtHR, 28 April 2009,
		App. No. 32881/04 (K.H.
		and others v. Slovakia)
	The right to access complaint	Articles 6 and 8 ECHR
	systems	• Article 23 Convention on
		Human Rights and
		Biomedicine
		• Article 4(2(c) Patients'
		Rights Directive
		• Article 6.5 WHO Europe
		Declaration on the Promotion
		of the Rights of Patients in
		Europe (1994)
		• Article 77-79 GDPR
	The right to harm	Articles 6 and 8 ECHR
	compensation	• Article 24 Convention on
		Human Rights and
		Biomedicine
		Article 82 GDPR

#### **5.5 Considerations for Implementation**

While codification has clear advantages for the protection of patients' rights, it is also important to consider the steps following the adoption of a new Charter: the implementation phase. National implementation is essential to supporting new rights (Bob, 2011). For example, the neurorights movement has been critiqued for having too little attention for the implementation. Some scholars have expressed concerns about the practical implications, suggesting the need for precise laws over new rights (Ligthart et al., 2023). Others found that enforcing existing rights is more important than creating new ones (Moreu Carbonell, 2021), or even warned of the devaluation of existing rights (Bublitz, 2022). Some argued that neurorights are legally unnecessary because existing frameworks already protect essential rights and freedoms such as the freedom of thought, consent, equality, and privacy (Borbón & Borbón, 2021). Similar critiques could be applied to (digital) patients' rights.

Indeed, mere formal recognition of such rights is only part of the solution. The codification in a legal instrument is not enough to effectively guarantee the protection of patients and thus cannot stand on its own. In parallel to the adoption of a Charter, it is therefore important to take concrete legal measures to further regulate the boundaries of AI technologies, as well as establish legal mechanisms for judicial protection of these rights, for example by adopting sector-specific rules for the marketing, development, and use of AI systems in healthcare in the context of the AI Act.

It is also worth noting that, naturally, this Charter would not exist in a legal vacuum, but instead enter a highly regulated scene, with many existing product regulations at the level of the EU, but also national laws and policies on medical professional conduct. Nevertheless, a Charter would provide welcome guidance on how to involve patients in a quickly changing medical landscape. In addition to this, it could be helpful to develop guidelines and tools for healthcare professionals on *how* to protect patients' rights when deploying AI technology in healthcare, such as in the choice of a specific AI system, for example in the form of a practical patients' rights assessment instrument (Van Kolfschooten, 2024a).

# 6. CONCLUDING REMARKS

The integration of AI in healthcare, while promising, introduces significant risks related to the health and rights of patients. The advancement of AI technologies in healthcare necessitates a balanced approach to integrating these innovations while safeguarding patients' rights. The proposed EU Charter for Digital Patients' Rights addresses the pressing need for a unified framework to protect patients in the age of AI. This Charter codifies existing rights and advocates for essential new rights. By harmonizing these rights across EU Member States, the Charter aims to mitigate health inequities and ensure that technological advancements enhance rather than undermine patient care. The implementation of such a Charter, coupled with effective regulatory mechanisms and guidelines for healthcare professionals, is crucial for maintaining trust in the evolving medical landscape. These efforts collectively aim to ensure that the integration of digital health innovation in healthcare enhances patient care without compromising human rights.

**Conflict of interest**: The corresponding author states that there is no conflict of interest. **Data Availability statement**: N/A

## **Reference list**

Abbing, H. D. C. R. (2010). Patients' Rights in a Technology and Market Driven-Europe. *European Journal of Health Law*, 17(1), 11–22.

https://doi.org/10.1163/157180909X12604572349566

- Agarwal, R., Bjarnadottir, M., Rhue, L., Dugas, M., Crowley, K., Clark, J., & Gao, G. (2022). Addressing algorithmic bias and the perpetuation of health inequities: An AI bias aware framework. *Health Policy and Technology*, 100702. https://doi.org/10.1016/j.hlpt.2022.100702
- Ahmed, M. I., Spooner, B., Isherwood, J., Lane, M., Orrock, E., & Dennison, A. (n.d.). A Systematic Review of the Barriers to the Implementation of Artificial Intelligence in Healthcare. *Cureus*, 15(10), e46454. https://doi.org/10.7759/cureus.46454

Alemanno, A. (2020). Towards a European Health Union: Time to Level Up. *European* Journal of Risk Regulation, 11(4), 721–725. https://doi.org/10.1017/err.2020.106

- Ali, R., Connolly, I. D., Tang, O. Y., Mirza, F. N., Johnston, B., Abdulrazeq, H. F., Lim, R. K., Galamaga, P. F., Libby, T. J., Sodha, N. R., Groff, M. W., Gokaslan, Z. L., Telfeian, A. E., Shin, J. H., Asaad, W. F., Zou, J., & Doberstein, C. E. (2024).
  Bridging the literacy gap for surgical consents: An AI-human expert collaborative approach. *Npj Digital Medicine*, 7(1), 1–6. https://doi.org/10.1038/s41746-024-01039-2
- Alston, P. (1984). Conjuring up New Human Rights: A Proposal for Quality Control. *The American Journal of International Law*, 78(3), 607–621. https://doi.org/10.2307/2202599
- Andorno, R., & Lavazza, A. (2023). How to deal with mind-reading technologies. *Frontiers in Psychology*, *14*. https://doi.org/10.3389/fpsyg.2023.1290478
- Bærøe, K., Miyata-Sturm, A., & Henden, E. (2020). How to achieve trustworthy artificial intelligence for health. *Bulletin of the World Health Organization*, 98(4), 257–262. https://doi.org/10.2471/BLT.19.237289
- Barbek, R., Toussaint, A., Löwe, B., & von dem Knesebeck, O. (2024). Intersectional inequalities in somatic symptom severity in the adult population in Germany found within the SOMA.SOC study. *Scientific Reports*, *14*(1), 3820. https://doi.org/10.1038/s41598-024-54042-8
- Bartlett, O. (2024). Can EU Law Be Used to Challenge Better Regulation Practices That Do Not Lead to Better Health? *European Journal of Health Law, 31*, 389-410. https://doi:10.1163/15718093-bja10129
- Bertolaso, M., & Marcos, A. (2022). Technological Changes and Rights Evolution in the Biodigital Era: A Philosophical Overview. In E. Stefanini, L. Liguori, M. Ienca, O.

Pollicino, & R. Andorno (Eds.), *The Cambridge Handbook of Information Technology, Life Sciences and Human Rights* (pp. 261–272). Cambridge University
Press. https://doi.org/10.1017/9781108775038.022

- Bester, J., Cole, C. M., & Kodish, E. (2016). The Limits of Informed Consent for an Overwhelmed Patient: Clinicians' Role in Protecting Patients and Preventing Overwhelm. *AMA Journal of Ethics*, *18*(9), 869–886. https://doi.org/10.1001/journalofethics.2016.18.9.peer2-1609
- Beyrer, C., Kamarulzaman, A., Isbell, M., Amon, J., Baral, S., Bassett, M. T., Cepeda, J.,
  Deacon, H., Dean, L., Fan, L., Giacaman, R., Gomes, C., Gruskin, S., Goyal, R., Mon,
  S. H. H., Jabbour, S., Kazatchkine, M., Kasoka, K., Lyons, C., ... Rubenstein, L.
  (2024). Under threat: The International AIDS Society–*Lancet* Commission on Health
  and Human Rights. *The Lancet*. https://doi.org/10.1016/S0140-6736(24)00302-7
- Blease, C., Torous, J., McMillan, B., Hägglund, M., & Mandl, K. D. (2024). Generative Language Models and Open Notes: Exploring the Promise and Limitations. *JMIR Medical Education*, 10, e51183. https://doi.org/10.2196/51183
- Blomberg, S. N., Folke, F., Ersbøll, A. K., Christensen, H. C., Torp-Pedersen, C., Sayre, M.
  R., Counts, C. R., & Lippert, F. K. (2019). Machine learning as a supportive tool to recognize cardiac arrest in emergency calls. *Resuscitation*, *138*, 322–329.
  https://doi.org/10.1016/j.resuscitation.2019.01.015
- Bob, C. (2011). Fighting for New Rights. In C. Bob (Ed.), *The International Struggle for New Human Rights* (pp. 1–13). University of Pennsylvania Press. https://doi.org/10.9783/9780812201345.1
- Borbón, D., & Borbón, L. (2021). A Critical Perspective on NeuroRights: Comments Regarding Ethics and Law. *Frontiers in Human Neuroscience*, 15, 703121. https://doi.org/10.3389/fnhum.2021.703121

- Borbón, D., & Ramírez-Gómez, J. A. (2024). Between politics and scholarship: The (un)settled debate over neurorights. *Frontiers in Political Science*, 6. https://doi.org/10.3389/fpos.2024.1335561
- Brent Mittelstadt. (2021). *The impact of Artificial Intelligence on the doctor-patient relationship*. Council of Europe.
- Broken links and bad advice: Conversations with SARAH, WHO's virtual health aide. (2024, April 23). POLITICO. https://www.politico.eu/article/bad-advice-conversations-with-sarah-world-health-organization-virtual-health-aide/
- Brown, C., Ruck Keene, A., Hooper, C. R., & O'Brien, A. (2020). Isolation of patients in psychiatric hospitals in the context of the COVID-19 pandemic: An ethical, legal, and practical challenge. *International Journal of Law and Psychiatry*, 71, 101572. https://doi.org/10.1016/j.ijlp.2020.101572
- Bublitz, J. C. (2022). Novel Neurorights: From Nonsense to Substance. *Neuroethics*, *15*(1), Article 1. https://doi.org/10.1007/s12152-022-09481-3
- Calcoen, P., Boer, A., & van de Ven, W. P. M. M. (n.d.). Should new health technology be available only for patients able and willing to pay? *Journal of Market Access and Health Policy*, 1(5), 1315294. https://doi.org/10.1080/20016689.2017.1315294
- Celeste, C., Ming, D., Broce, J., Ojo, D. P., Drobina, E., Louis-Jacques, A. F., Gilbert, J. E., Fang, R., & Parker, I. K. (2023). Ethnic disparity in diagnosing asymptomatic bacterial vaginosis using machine learning. *Npj Digital Medicine*, 6(1), 1–10. https://doi.org/10.1038/s41746-023-00953-1
- Chalabi, A. (2023). A New Theoretical Model of the Right to Environment and its Practical Advantages. *Human Rights Law Review*, 23(4), ngad023. https://doi.org/10.1093/hrlr/ngad023

Cocito, C., & De Hert, P. (2023). The transformative nature of the EU Declaration on Digital Rights and Principles: Replacing the old paradigm (normative equivalency of rights). *Computer Law & Security Review*, 50, 105846.
https://doi.org/10.1016/j.clsr.2023.105846

Cohen, J., & Ezer, T. (2013). Human rights in patient care: A theoretical and practical framework. *Health and Human Rights Journal*. https://www.hhrjournal.org/2013/12/human-rights-in-patient-care-a-theoretical-andpractical-framework/

Crenshaw, K. (1989). Demarginalizing the Intersection of Race and Sex: A Black Feminist Critique of Antidiscrimination Doctrine, Feminist Theory and Antiracist Politics. U. Chi. Legal F., 1989, 139.

https://scholarship.law.columbia.edu/faculty\_scholarship/3007

Custers, B. (2022). New digital rights: Imagining additional fundamental rights for the digital era. *Computer Law & Security Review*, 44, 105636. https://doi.org/10.1016/j.clsr.2021.105636

Daigle, B., & Khan, M. (2020). The EU General Data Protection Regulation: An Analysis of Enforcement Trends by EU Data Protection Authorities. *Journal of International Commerce & Economics*, 2020, 1. https://heinonline.org/HOL/Page?handle=hein.journals/jice2020&id=1&div=&collecti on=

- Davies, B. (2020). The right not to know and the obligation to know. *Journal of Medical Ethics*, 46(5), 300–303. https://doi.org/10.1136/medethics-2019-106009
- De Ruijter, A. (2019). EU Health Law & Policy: The Expansion of EU Power in Public Health and Health Care. Oxford University Press.

- de Witte, B. (2006). Non-market values in internal market legislation. In Shuibhne, Nic Niamh (Ed.), *Regulating the Internal Market* (pp. 61–86). Edward Elgar.
- Delhomme, V. (2017). Internal Market as an Excuse: The Case of EU Anti-Tobacco Legislation. College of Europe.
- Delhomme, V. (2020). Emancipating Health from the Internal Market: For a Stronger EU (Legislative) Competence in Public Health. *European Journal of Risk Regulation*, 11(4), 747–756. https://doi.org/10.1017/err.2020.85
- DELVAUX, M. (n.d.). *REPORT with recommendations to the Commission on Civil Law Rules on Robotics* | *A8-0005/2017* | *European Parliament*. Retrieved July 3, 2024, from https://www.europarl.europa.eu/doceo/document/A-8-2017-0005\_EN.html
- Duffourc, M. N., & Gerke, S. (2023). The proposed EU Directives for AI liability leave worrying gaps likely to impact medical AI. *Npj Digital Medicine*, 6(1), Article 1. https://doi.org/10.1038/s41746-023-00823-w
- Durán, J. M., & Jongsma, K. R. (2021). Who is afraid of black box algorithms? On the epistemological and ethical basis of trust in medical AI. *Journal of Medical Ethics*, 47(5), 329–335. https://doi.org/10.1136/medethics-2020-106820
- Finco, M. G., Mir, N., Gresham, G., & Huisingh-Scheetz, M. (2024). Ethical considerations of digital health technology in older adult care. *The Lancet Healthy Longevity*, 5(1), e12–e13. https://doi.org/10.1016/S2666-7568(23)00236-2
- Foundations of Global Health & Human Rights. (2020). Oxford University Press. https://doi.org/10.1093/oso/9780197528297.001.0001
- Ghazarian, A., Zheng, J., El-Askary, H., Chu, H., Fu, G., & Rakovski, C. (2021). Increased
  Risks of Re-identification For Patients Posed by Deep Learning-Based ECG
  Identification Algorithms. 2021 43rd Annual International Conference of the IEEE

Engineering in Medicine & Biology Society (EMBC), 1969–1975. https://doi.org/10.1109/EMBC46164.2021.9630880

- Greenwood, B. N., Carnahan, S., & Huang, L. (2018). Patient–physician gender concordance and increased mortality among female heart attack patients. *Proceedings of the National Academy of Sciences*, *115*(34), 8569–8574. https://doi.org/10.1073/pnas.1800097115
- Greer, S. L., Rozenblum, S., Fahy, N., Brooks, E., Jarman, H., de Ruijter, A., Palm, W., &
  Wismar, M. (2022a). *Everything you always wanted to know about European Union health policies but were afraid to ask*. World Health Organization. Regional Office for
  Europe.
- Greer, S. L., Rozenblum, S., Fahy, N., Brooks, E., Jarman, H., de Ruijter, A., Palm, W., &
  Wismar, M. (2022b). *Everything you always wanted to know about European Union health policies but were afraid to ask*. World Health Organization. Regional Office for
  Europe.
- Grodin, M. A., Tarantola, D., Annas, G. J., & Gruskin, S. (Eds.). (2013). Health and Human Rights in a Changing World. Routledge. https://www.routledge.com/Health-and-Human-Rights-in-a-Changing-World/Grodin-Tarantola-Annas-Gruskin/p/book/9780415503990

Hefti, A. (2024). Intersectional Victims as Agents of Change in International Human Rights-Based Climate Litigation. *Transnational Environmental Law*, 1–26. https://doi.org/10.1017/S2047102524000128

Hermes, S., Riasanow, T., Clemons, E. K., Böhm, M., & Krcmar, H. (2020). The digital transformation of the healthcare industry: Exploring the rise of emerging platform ecosystems and their influence on the role of patients. *Business Research*, 13(3), 1033–1069. https://doi.org/10.1007/s40685-020-00125-x

- Hertz, N. (2022). Neurorights Do we Need New Human Rights? A Reconsideration of the Right to Freedom of Thought. *Neuroethics*, 16(1), 5. https://doi.org/10.1007/s12152-022-09511-0
- Hervey, T. K., & McHale, J. V. (2015). European Union Health Law: Themes and Implications. Cambridge University Press.
- Ienca, M. (2021). On Neurorights. Frontiers in Human Neuroscience, 15, 701258. https://doi.org/10.3389/fnhum.2021.701258

Ienca, M., & Andorno, R. (2017). Towards new human rights in the age of neuroscience and neurotechnology. *Life Sciences, Society and Policy*, 13(1), 5. https://doi.org/10.1186/s40504-017-0050-1

- #InternetOfRights: Creating the Universal Declaration of Digital Rights. (2017, March 24). ARTICLE 19. https://www.article19.org/resources/internetofrights-creating-theuniversal-declaration-of-digital-rights/
- Johnson, W. G. (2024). Beyond Substance: Structural and Political Questions for Neurotechnologies and Human Rights. AJOB Neuroscience, 15(2), 134–136. https://doi.org/10.1080/21507740.2024.2326915
- Kapilashrami, A., & Aziz, R. (2023). Pandemic preparedness with 20/20 vision: Applying an intersectional equity lens to health workforce planning. *The International Journal of Health Planning and Management*, 38(5), 1117–1126.
  https://doi.org/10.1002/hpm.3677
- Khawaja, Z., & Bélisle-Pipon, J.-C. (2023). Your robot therapist is not your therapist:
  Understanding the role of AI-powered mental health chatbots. *Frontiers in Digital Health*, 5, 1278186. https://doi.org/10.3389/fdgth.2023.1278186
- Kickbusch, I., Piselli, D., Agrawal, A., Balicer, R., Banner, O., Adelhardt, M., Capobianco,E., Fabian, C., Gill, A. S., Lupton, D., Medhora, R. P., Ndili, N., Ryś, A., Sambuli, N.,

Settle, D., Swaminathan, S., Morales, J. V., Wolpert, M., Wyckoff, A. W., ... Wong, B. L. H. (2021). The Lancet and Financial Times Commission on governing health futures 2030: Growing up in a digital world. *The Lancet*, *398*(10312), 1727–1776. https://doi.org/10.1016/S0140-6736(21)01824-9

- Kim, T. W., & Routledge, B. R. (2022). Why a Right to an Explanation of Algorithmic Decision-Making Should Exist: A Trust-Based Approach. *Business Ethics Quarterly*, 32(1), 75–102. https://doi.org/10.1017/beq.2021.3
- Krakowski, I., Kim, J., Cai, Z. R., Daneshjou, R., Lapins, J., Eriksson, H., Lykou, A., & Linos, E. (2024). Human-AI interaction in skin cancer diagnosis: A systematic review and meta-analysis. *Npj Digital Medicine*, 7(1), 1–10. https://doi.org/10.1038/s41746-024-01031-w
- Laitinen, Arto, Marketta, Niemela, & Pirhonen, J. (2016). Social robotics, elderly care, and human dignity: A recognition-theoretical approach. In Seibt, J. (Ed.), *What Social Robots Can and Should Do*. IOS Press.

https://lib.uva.nl/discovery/fulldisplay/cdi\_scopus\_primary\_2\_s2\_0\_84992549249/31 UKB\_UAM1\_INST:UVA

Lau, P. L. (n.d.). AI Gender Biases in Women's Healthcare: Perspectives from the United Kingdom and the European Legal Space (pp. 1–28). Springer International Publishing. https://doi.org/10.1007/16495\_2023\_63

Legemaate, J. (2024). *Beter (met) gezondheidsrecht*. Boom Juridisch. https://www.scheltema.nl/boek?authortitle=johan-legemaate/beter-met-gezondheidsrecht--9789462129504

Leslie, D., Mazumder, A., Peppin, A., Wolters, M. K., & Hagerty, A. (2021). Does "AI" stand for augmenting inequality in the era of covid-19 healthcare? *BMJ*, 372, n304. https://doi.org/10.1136/bmj.n304

- Ligthart, S., Bublitz, C., Douglas, T., Forsberg, L., & Meynen, G. (2022). Rethinking the Right to Freedom of Thought: A Multidisciplinary Analysis. *Human Rights Law Review*, 22(4), ngac028. https://doi.org/10.1093/hrlr/ngac028
- Ligthart, S., Ienca, M., Meynen, G., Molnar-Gabor, F., Andorno, R., Bublitz, C., Catley, P., Claydon, L., Douglas, T., Farahany, N., Fins, J. J., Goering, S., Haselager, P., Jotterand, F., Lavazza, A., McCay, A., Wajnerman Paz, A., Rainey, S., Ryberg, J., & Kellmeyer, P. (2023). Minding Rights: Mapping Ethical and Legal Foundations of 'Neurorights.' *Cambridge Quarterly of Healthcare Ethics*, *32*(4), Article 4. https://doi.org/10.1017/S0963180123000245
- Logan, J., Kennedy, P. J., & Catchpoole, D. (2023). A review of the machine learning datasets in mammography, their adherence to the FAIR principles and the outlook for the future. *Scientific Data*, *10*(1), 595. https://doi.org/10.1038/s41597-023-02430-6
- London, L. (2008). What is a human-rights based approach to health and does it matter? *Health and Human Rights*, 10(1), 65–80.
- López-Silva, P., Wajnerman-Paz, A., & Molnar-Gabor, F. (2024). Neurotechnological Applications and the Protection of Mental Privacy: An Assessment of Risks. *Neuroethics*, *17*(2), 31. https://doi.org/10.1007/s12152-024-09565-2
- Mann, J. M., Grodin, M. A., Gruskin, S., & Annas, G. J. (Eds.). (1999). *Health and Human Rights: A Reader* (First Edition). Routledge.
- McHale, J. V., & Hervey, T. K. (Eds.). (2015). Rights: Health rights as human rights. In European Union Health Law: Themes and Implications (pp. 156–183). Cambridge University Press. https://doi.org/10.1017/CBO9780511862410.008
- Medrano, D., Matsumoto, M. M., Flores, E., & Dako, F. (2023). Understanding Race-Based Medicine and Its Impact on Radiology. *Journal of the American College of Radiology*, 20(7), 623–628. https://doi.org/10.1016/j.jacr.2023.03.017

- Mira, J. J., Lorenzo, S., Guilabert, M., & Pérez-Jover, V. (2012). Do Spaniards know their rights as patients? *International Journal for Quality in Health Care*, 24(4), 365–370. https://doi.org/10.1093/intqhc/mzs025
- Montemayor, C., Halpern, J., & Fairweather, A. (2022). In principle obstacles for empathic AI: Why we can't replace human empathy in healthcare. *AI & SOCIETY*, *37*(4), 1353– 1359. https://doi.org/10.1007/s00146-021-01230-z
- Moreu Carbonell, E. (2021). The Regulation of Neuro-Rights\*. *European Review of Digital Administration & Law*, 2(2), 149–162.
- Murphy, T. (2009). Repetition, Revolution, and Resonance: An Introduction to New Technologies and Human Rights. In T. Murphy (Ed.), *New Technologies and Human Rights* (p. 0). Oxford University Press.

https://doi.org/10.1093/acprof:oso/9780199562572.003.0001

Murphy, T. (2013). Health and Human Rights: 18.

- Nakagawa, K., Moukheiber, L., Celi, L. A., Patel, M., Mahmood, F., Gondim, D., Hogarth, M., & Levenson, R. (2023). AI in Pathology: What could possibly go wrong?
  Seminars in Diagnostic Pathology, 40(2), 100–108.
  https://doi.org/10.1053/j.semdp.2023.02.006
- Nawrot, O., Szudejko, P., & Vachev, V. (2024). The interior castle of conscience vs. New human rights in the age of neuroscience and neurotechnology. *The International Journal of Human Rights*, *28*(2), 177–196.

https://doi.org/10.1080/13642987.2023.2234301

Naylor, B., & Shalev, S. (2022). Solitary Confinement and the Meaning of 'Meaningful Human Contact.' In C. Burbano Herrera & Y. Haeck (Eds.), *Human Rights Behind Bars: Tracing Vulnerability in Prison Populations Across Continents from a* *Multidisciplinary Perspective* (pp. 293–318). Springer International Publishing. https://doi.org/10.1007/978-3-031-11484-7 13

- Nys, H. (2001). Comparative health law and the harmonization of patients' rights in Europe. *European Journal of Health Law*, 8(4), 317–331. https://www.jstor.org/stable/45442515
- Nys, H., & Goffin, T. (2011). Mapping national practices and strategies relating to patients' rights. In *Cross-border Health Care in the European Union Mapping and analysing practices and policies*.
- Ó Cathaoir, K. (2022). Digital Healthcare Technologies and Human Rights. In *Health and human rights: Global and European perspectives* (2nd edition). Intersentia.

OECD. (2024). AI in Health: Huge Potential, Huge Risks. OECD.

Outcome Document of the First Meeting of the AHEG: First Draft of a Recommendation on the Ethics of Neurotechnology (First Version), SHS/BIO/AHEG-Neuro/2024/1 Rev., UNESCO Digital Library (2024).

https://unesdoc.unesco.org/ark:/48223/pf0000389768?posInSet=1&queryId=cfc56e24 -7a88-4b19-90e7-63bcfc4e556b

- Palm, W., Nys, H., Townend, D., Shaw, D., Clemens, T., & Brand, H. (2020). Patients' rights: From recognition to implementation. In A. Anell, E. Nolte, & S. Merkur (Eds.), *Achieving Person-Centred Health Systems: Evidence, Strategies and Challenges* (pp. 347–386). Cambridge University Press. https://doi.org/10.1017/9781108855464.016
- Ploem, C. (2023). *Goed recht voor zorgtechnologie*. Boom Juridisch. https://www.boom.nl/auteur/110-12813\_Ploem/100-14613\_Goed-recht-voorzorgtechnologie

- Ploug, T., & Holm, S. (2020a). The four dimensions of contestable AI diagnostics—A patient-centric approach to explainable AI. *Artificial Intelligence in Medicine*, 107, 101901. https://doi.org/10.1016/j.artmed.2020.101901
- Ploug, T., & Holm, S. (2020b). The right to refuse diagnostics and treatment planning by artificial intelligence. *Medicine, Health Care and Philosophy*, 23(1), 107–114. https://doi.org/10.1007/s11019-019-09912-8
- Poirot, M. G., Ruhe, H. G., Mutsaerts, H.-J. M. M., Maximov, I. I., Groote, I. R., Bjørnerud, A., Marquering, H. A., Reneman, L., & Caan, M. W. A. (2024). Treatment Response Prediction in Major Depressive Disorder Using Multimodal MRI and Clinical Data: Secondary Analysis of a Randomized Clinical Trial. *American Journal of Psychiatry*, 181(3), 223–233. https://doi.org/10.1176/appi.ajp.20230206
- Pollicino, O. (2022). The Right to Internet Access: A Comparative Constitutional Legal Framework. In E. Stefanini, L. Liguori, M. Ienca, O. Pollicino, & R. Andorno (Eds.), *The Cambridge Handbook of Information Technology, Life Sciences and Human Rights* (pp. 125–138). Cambridge University Press. https://doi.org/10.1017/9781108775038.013
- Postigo, H. (2012). *The digital rights movement: The role of technology in subverting digital copyright*. MIT Press.
- Prince, A. E. R., & Schwarz, D. (2020). Proxy Discrimination in the Age of Artificial Intelligence and Big Data. *Iowa Law Review*, 1257(105).
  https://ilr.law.uiowa.edu/print/volume-105-issue-3/proxy-discrimination-in-the-age-ofartificial-intelligence-and-big-data
- Puri, A. (2023). The Group Right to Mutual Privacy. *Digital Society*, *2*(2), 22. https://doi.org/10.1007/s44206-023-00051-5

Purnhagen, K. P., Ruijter, A. D., Flear, M. L., Hervey, T. K., & Herwig, A. (2020). More Competences than You Knew? The Web of Health Competence for European Union Action in Response to the COVID-19 Outbreak. *European Journal of Risk Regulation*, 11(2), 297–306. https://doi.org/10.1017/err.2020.35

Rodríguez, V., Andrade, A. D., García-Retamero, R., Anam, R., Rodríguez, R., Lisigurski,
M., Sharit, J., & Ruiz, J. G. (2013). Health literacy, numeracy, and graphical literacy
among veterans in primary care and their effect on shared decision making and trust in
physicians. *Journal of Health Communication*, *18 Suppl 1*(Suppl 1), 273–289.
https://doi.org/10.1080/10810730.2013.829137

- Rothmar Herrmann, Janne, Katharina, Ó. C., & Toebes, Brigit. (2022). The European Union: A Health and Human Rights Perspective. In *Health and human rights: Global and European perspectives* (2nd edition). Intersentia.
- Ruijter, A. de. (2016). European integration in the field of human health. *Journal of European Integration*, *38*(7), 837–843. https://doi.org/10.1080/07036337.2016.1228333
- Ruijter, A. de. (2019). EU Health Law & Policy: The Expansion of EU Power in Public Health and Health Care.
- Sauter, W., & Guy, M. (2017). The history and scope of EU health law and policy. In T. K. Hervey, C. A. Young, & L. E. Bishop (Eds.), *Research Handbook on EU Health Law* and Policy (pp. 17–35). Edward Elgar Publishing. https://doi.org/10.2139/ssrn.2718045
- Schwarcz, D. (2021). Health-Based Proxy Discrimination, Artificial Intelligence, and Big Data Symposium Articles. *Houston Journal of Health Law & Policy*, 21(1), 95–124. https://heinonline.org/HOL/P?h=hein.journals/hhpol21&i=107

Shaw, R. J. (2023). Access to Technology and Digital Literacy as Determinants of Health and Health Care. *Creative Nursing*, 29(3), 258–263. https://doi.org/10.1177/10784535231211682

Shen, F. X. (2013). Neuroscience, mental privacy, and the law. Harvard Journal of Law & Public Policy, 36(2), 653–680. https://go-galecom.proxy.uba.uva.nl/ps/i.do?p=AONE&sw=w&issn=01934872&v=2.1&it=r&id=G ALE%7CA330143494&sid=googleScholar&linkaccess=abs

Takshi, S. (2020). Unexpected Inequality: Disparate-Impact from Artificial Intelligence in Healthcare Decisions. *Journal of Law and Health*, 34, 215. https://heinonline.org/HOL/Page?handle=hein.journals/jlah34&id=222&div=&collecti on=

Toebes, B. (2021). Socioeconomic Health Inequalities in Europe: The Role of Law and Human Rights. In D. Orentlicher & T. K. Hervey (Eds.), *The Oxford Handbook of Comparative Health Law* (pp. 171–187). Oxford University Press. https://doi.org/10.1093/oxfordhb/9780190846756.013.44

Toebes, B. C. A., Hartlev, M., Hendriks, A., Ó Cathaoir, K., Herrmann, J. R., & Aasen, H. S. (2022). *Health and human rights: Global and European perspectives* (2nd edition.). Intersentia.

Townend, D., Clemens, T., Shaw, D., Brand, H., Nys, H., & Palm, W. (2018). Patients' rights in the European Union: Mapping eXercise: Final report—Study. European Commission.

Tsai, T.-I., Yu, W.-R., & Lee, S.-Y. D. (2018). Is health literacy associated with greater medical care trust? *International Journal for Quality in Health Care: Journal of the International Society for Quality in Health Care*, 30(7), 514–519. https://doi.org/10.1093/intqhc/mzy043

- Tudorie, G. (2023). Reluctant Republic: A Positive Right for Older People to Refuse AI-Based Technology. *Societies*, *13*(12), Article 12. https://doi.org/10.3390/soc13120248
- UNESCO. (2023). *The risks and challenges of neurotechnologies for human rights*. UNESCO. https://unesdoc.unesco.org/ark:/48223/pf0000384185

van Kolfschooten, H. (2022a). EU regulation of artificial intelligence: Challenges for patients' rights. *Common Market Law Review*, 59(1). https://kluwerlawonline.com/journalarticle/Common+Market+Law+Review/59.1/COL A2022005

- van Kolfschooten, H. (2022b). The mHealth Power Paradox: Improving Data Protection in Health Apps through Self-Regulation in the European Union. In I. G. Cohen, T. Minssen, W. N. Price II, C. Robertson, & C. Shachar (Eds.), *The Future of Medical Device Regulation: Innovation and Protection* (pp. 63–76). Cambridge University Press. https://doi.org/10.1017/9781108975452.006
- van Kolfschooten, H. (2023). The AI cycle of health inequity and digital ageism: Mitigating biases through the EU regulatory framework on medical devices. *Journal of Law and the Biosciences*, *10*(2), lsad031. https://doi.org/10.1093/jlb/lsad031
- van Kolfschooten, H. (2024a). A Human Rights-Based Approach to Artificial Intelligence in Healthcare: A Proposal for a Patients' Rights Impact Assessment Tool (SSRN Scholarly Paper 4861569). SSRN. https://doi.org/10.2139/ssrn.4861569
- van Kolfschooten, H. (2024b). A health-conformant reading of the GDPR's right not to be subject to automated decision-making. *Medical Law Review*, 32(3), 373–391. <u>https://doi.org/10.1093/medlaw/fwae029</u>
- van Kolfschooten, H., & van Oirschot, J. (2024) The EU Artificial Intelligence Act (2024): Implications for healthcare. *Health Policy*, 149, Article 105152. https://doi.org/10.1016/j.healthpol.2024.105152

- van Kolfschooten, H., & Shachar, C. (2023). The Council of Europe's AI Convention (2023–2024): Promises and pitfalls for health protection. *Health Policy*, *138*, 104935. https://doi.org/10.1016/j.healthpol.2023.104935
- Veale, M., & Borgesius, F. Z. (2021). Demystifying the Draft EU Artificial Intelligence Act. *Computer Law Review International*, 22(4), 97–112. https://doi.org/10.9785/cri-2021-220402
- von Arnauld, A., von der Decken, K., & Susi, M. (Eds.). (2020). The Cambridge Handbook of New Human Rights: Recognition, Novelty, Rhetoric. Cambridge University Press. https://doi.org/10.1017/9781108676106

von der Decken, K., & Koch, N. (2020). Recognition of New Human Rights: Phases, Techniques and the Approach of 'Differentiated Traditionalism.' In A. von Arnauld, K. von der Decken, & M. Susi (Eds.), *The Cambridge Handbook of New Human Rights: Recognition, Novelty, Rhetoric* (pp. 7–20). Cambridge University Press. https://doi.org/10.1017/9781108676106.002

- Weber, F. (2022). Cross-border cooperation in the border region of Germany, France, and Luxembourg in times of Covid-19. *European Societies*, 24(3), 354–381. https://doi.org/10.1080/14616696.2022.2076894
- World Health Organization. (2023). *Regulatory considerations on artificial intelligence for health*. World Health Organization.

Wynants, L., Calster, B. V., Collins, G. S., Riley, R. D., Heinze, G., Schuit, E., Albu, E.,
Arshi, B., Bellou, V., Bonten, M. M. J., Dahly, D. L., Damen, J. A., Debray, T. P. A.,
Jong, V. M. T. de, Vos, M. D., Dhiman, P., Ensor, J., Gao, S., Haller, M. C., ...
Smeden, M. van. (2020). Prediction models for diagnosis and prognosis of covid-19:
Systematic review and critical appraisal. *BMJ*, *369*, m1328.
https://doi.org/10.1136/bmj.m1328

Yi, F., Zhang, Y., Yuan, J., Liu, Z., Zhai, F., Hao, A., Wu, F., Somekh, J., Peleg, M., Zhu, Y.-C., & Huang, Z. (2023). Identifying underlying patterns in Alzheimer's disease trajectory: A deep learning approach and Mendelian randomization analysis. *eClinicalMedicine*, 64. https://doi.org/10.1016/j.eclinm.2023.102247